Author's response to reviews

Title: Analysis of genetic copy number changes in cervical disease progression

Authors:

Frank A Policht (Frank.Policht@abbott.com)
Minghao Song (Minghao.Song@abbott.com)
Svetlana Sitailo (Svetlana.Sitailo@abbott.com)
Anna O'Hare (Anna.OHare@abbott.com)
Raheela Ashfaq (rashfaq@carisdx.com)
Carolyn Y Muller (cmuller@salud.unm.edu)
Larry E Morrison (lemorrison73@gmail.com)
Walter King (genedocw@comcast.net)
Irina A Sokolova (Irina.Sokolova@abbott.com)

Version: 3 Date: 11 August 2010

Author's response to reviews: see over
August 11, 2010

To Whom It May Concern:

Enclosed is a revised edition of the manuscript entitled “Analysis of Genetic Copy Number Changes in Cervical Disease Progression.”

The following changes were implemented in the manuscript:

1) The list of authors was aligned with those as listed in the submission system.
2) Changed the format of the authors’ contributions section from a list into a paragraph form and added a statement indicating that all authors read and approved the final manuscript.
3) Changed the case of the title in the manuscript file and in the submission system.
4) Added country information for authors in the author details section.
5) Removed the running title and keywords from the title page.
6) Removed vertical lines from Tables 1 and 2.
7) The figure numbers (titles) were removed from all figures.
8) The reference list was updated to list all authors in place of et al.

Based on these revisions, please accept it as a candidate for publication in BMC Cancer. This article is our original, unpublished work and it has not been submitted to any other journal for reviews.

Thank you for your consideration.

Sincerely,

Frank Policht